Author's response to reviews

Title: Trial on Refinement of Early stage non-small cell lung cancer. Adjuvant chemotherapy with pemetrexed and cisplatin versus vinorelbine and cisplatin: The TREAT protocol

Authors:

Michael Kreuter MK (michael.kreuter@thoraxklinik-heidelberg.de)
Johan Vansteenkiste JV (johan.vansteenkiste@uz.kuleuven.ac.be)
Frank Griesinger FG (fgriesi@gwdg.de)
Hans Hoffmann HH (hans.hoffmann@thoraxklinik-heidelberg.de)
Hendrik Dienemann HD (hendrik.dienemann@thoraxklinik-heidelberg.de)
Paul De Leyn PD (paul.deleyn@uz.kuleuven.ac.be)
Michael Thomas MT (michael.thomas@thoraxklinik-heidelberg.de)

Version: 2 Date: 7 May 2007

Author's response to reviews: see over
Re: Manuscript BMC cancer 3399932531387136

Dear Dr Saltman and Dr. Norton,

We would like to thank your for the acceptance of our manuscript entitled „Trial on Refinement of Early stage non-small cell lung cancer. Adjuvant chemotherapy with pemetrexed and cisplatin versus vinorelbine and cisplatin: The TREAT protocol”.

According to the requirements of the editorial production team, the following changes in formatting have been made:

1) Title: removal of the registration number
2) Formatting of the Trial Registration in the Abstract section, p. 2
3) Section headings were formatted for “Abstract”, p. 2 and for “Premature Withdrawal”, p. 18
4) “List of Abbreviations” was changed to “Abbreviations”, p. 23

We hope that all necessary changes have been made. For further editorial correspondence please refer to our Email address: michael.kreuter@thoraxklinik-heidelberg.de

With kindest regards,

Michael Kreuter, M.D.